Using Artificial Neural Networks to Couple Satellite C-Band Synthetic Aperture Radar Interferometry and Alpine3D Numerical Model for the Estimation of Snow Cover Extent, Height, and Density

Gianluca Palermo^(D), Edoardo Raparelli, Paolo Tuccella^(D), Massimo Orlandi^(D), and Frank Silvio Marzano^(D), *Fellow, IEEE*

Abstract—This work presents a new approach for the estimation of snow extent, height, and density in complex orography regions, which combines differential interferometric synthetic-apertureradar (DInSAR) data and snowpack numerical model data through artificial neural networks (ANNs). The estimation method, subdivided into classification and estimation, is based on two ANNs trained by a DInSAR response model coupled with Alpine3D snow cover numerical model outputs. Auxiliary satellite training data from satellite visible-infrared MODIS imager as well as digital elevation and land cover models are used to discriminate wet and dry snow areas. For snow cover classification the ANN-based estimation methodology is combined with fuzzy-logic and compared with a consolidated decision threshold approach using C-band SAR backscattering information. For snow height (SH) and density estimation, the proposed methodology is compared with an analytical inverse method and two model-based statistical techniques (linear regression and maximum likelihood). The validation is carried out in Central Apennines, a mountainous area in Italy with an extension of about 104 km² and peaks up to 2912 m, using in situ data collected between December 2018 and February 2019. Results show that the ANN-based technique has a snow cover area classification accuracy of more than 80% when compared MODIS maps. Estimation bias and root mean square error are equal to about 0.5 cm and 20 cm for SH and to 5 kg/m³ and 80 kg/m³ for snow density. As expected,

Manuscript received 15 March 2022; revised 15 September 2022 and 20 December 2022; accepted 18 February 2023. Date of publication 10 March 2023; date of current version 29 March 2023. This work was supported by the Sapienza University of Rome and CETEMPS and in part by the Agenzia Spaziale Italiana through the Project SMIVIA (Contract no. 2021-9-U.0 CUP F85F21001230005). (Corresponding author: Gianluca Palermo.)

Gianluca Palermo is with the Dipartimento di Ingegneria dell'Informazione, Elettronica e Telecomunicazioni, Sapienza Università di Roma, 00184 Rome, Italy (e-mail: gianluca.palermo@uniroma1.it).

Edoardo Raparelli is with the Dipartimento di Ingegneria dell'Informazione, Elettronica e Telecomunicazioni, Sapienza Università di Roma, 00184 Rome, Italy, and also with the Center of Excellence CETEMPS, Università dell'Aquila, 67100 L'Aquila, Italy (e-mail: edoardo.raparelli@uniroma1.it).

Paolo Tuccella is with the Department DSFC and the Center of Excellence CETEMPS, Università dell'Aquila, 67100 L'Aquila, Italy (e-mail: paolo.tuccella@aquila.infn.it).

Massimo Orlandi is with the Progressive Systems for ESA, 00044 Frascati, Italy (e-mail: orlandi.massimo@gmail.com).

Frank Silvio Marzano, deceased, was with the Dipartimento di Ingegneria dell'Informazione, Elettronica e Telecomunicazioni, Sapienza Università di Roma, 00184 Rome, Italy, and also with the Center of Excellence CETEMPS, Università dell'Aquila, 67100 L'Aquila, Italy (e-mail: frank.marzano@uniroma1.it).

Digital Object Identifier 10.1109/JSTARS.2023.3253804

worse results are associated with low DInSAR coherence between two repeat passes and snow melting periods.

Index Terms—Data fusion, differential interferometry, inversion methods, neural networks, snow cover modeling, snow cover retrieval, synthetic aperture radar (SAR).

I. INTRODUCTION

C EASONAL snow cover is the largest cryospheric component, covering more than 50 million square kilometers of the Earth surface (more than 31% of its land area) every year [1]. Snow cover area (SCA), snow height (SH), and snow density (SD) as well as snow water equivalent (SWE) are the main parameters characterizing the snow accumulation in mountainous regions. Snow cover patterns are governed by the effects of topography, land cover, wind redistribution, solar irradiance, and air temperature. Snow mapping is particularly important in meteorology, hydrology, and climate monitoring applications [2]. The considerable geographical extension of snow cover and its spatial heterogeneity makes it impractical to monitor the above parameters regularly (i.e., with a high spatial and temporal resolution) by means of direct or indirect in situ measurements. Moreover, in the last few decades, a general back-scaling of snow observation networks occurred worldwide [3]. The exploitation of satellite technologies can significantly help systematic snow monitoring [3]. Among the spaceborne remote sensing systems, synthetic aperture radar (SAR) instruments are particularly suitable for the analysis of snow deposits, providing data with spatial resolutions down to some meters, a global scale coverage and a few days revisit time [5], [6]. The centimetric wavelength of radar signal, capable of penetrating into snow layers, makes SAR spaceborne sensors particularly suitable for cryospheric applications [7], [8], [9], [10], [11].

The power and phase information at different frequencies and polarizations can be obtained from the specific processing of SAR images. Such information can be used to retrieve snow parameters and develop active-passive physical-electromagnetic response models. SAR backscattered power strongly depends on the air-snow and snow-ground interface reflections, as well as on the snow volumetric effects [8], [9]. Differential interferometric SAR (DInSAR) techniques have been successfully employed for

This work is licensed under a Creative Commons Attribution 4.0 License. For more information, see https://creativecommons.org/licenses/by/4.0/

SWE retrieval, exploiting the phase shift between two sequential images and taking into account the effects of air-snow refraction and the snow-ground reflection [25], [26], [27], [28], [29], [30], [31]. The repeat-pass SAR interferometry can provide a high-resolution mapping of snow depth, but it requires a good coherence between two passes [27], [31]. These limitations for snow detection and mapping may be approached by combining SAR microwave observations with satellite optical spectrometer data at high-moderate spatial resolution [33], [34], [35], [36]. Spaceborne SAR polarimetry can also be exploited since the interaction between the polarized incident wave and the snow anisotropic features causes depolarization effects and specific scattering mechanisms [37], [38], [39], [40], [41]. Satellitebased passive microwave measurements may also be useful in combination with SAR observations, even though with a spatial resolution (tens of kilometers) much worse than the SAR typical resolutions [42], [43].

Most snow remote sensing applications from space have been focused on major mountain systems, such as Sierra Nevada, Alps, and Himalayan regions [14], [15], [29], [31]. The Italian Apennines, have never been considered for this kind of studies by the scientific community, probably due to the high spatial and temporal variability of their snow cover [44]. Nevertheless, the central Apennines region holds a key role for the meteorological dynamics in the Mediterranean area and, moreover, it hosts the southernmost European glacier, the Calderone. The latter is to be considered a glacieret rather than a glacier and its evolution represents a relevant indicator, at least for the medium latitudes, of the impact of climatic changes [45]. In such complex geographical regions, snow cover numerical models, coupled with meteorological weather forecast, can also be exploited to characterize the microphysics and dynamics of the snow cover [46], [47]. Snow cover model simulations can be used to provide auxiliary information in the satellite-based retrieval of the snow cover [48]. On the other hand, meteorological model predictions can be used to mitigate atmospheric water vapor effects on SAR interferograms [49], [50], [51], [52].

In this work, we present a DInSAR retrieval methodology to estimate SCA, height, and density in complex orography regions using Sentinel-1 satellite C-band data. The retrieval methodology is subdivided into classification and estimation subsequent stages. Classification is based on artificial neural networks (ANNs) and fuzzy logic. Estimation is based on an ANN trained by a forward DInSAR response model, coupled with snow cover numerical model outputs. In order to discriminate between wet and dry snow areas, data from satellite visibleinfrared imager are used. A digital elevation model (DEM) and a land cover database are also used to assist in the discrimination process. For snow cover classification, the results obtained from ANN-based retrieval methodology are compared with results from well-established decision threshold approaches. For SH and density estimation, the proposed methodology is compared, in terms of the obtained results, with the well-known analytical inverse method and two model-based statistical techniques (linear regression and maximum likelihood). The validation is carried out in the Italian Central Apennines, using in situ snow data collected between December 2018 and February 2019.

Results show that the ANN-based technique has a relatively good performance in terms of SCA classification accuracy, together with an estimation bias of about 0.5 cm and 5 kg/m³ for SH and density, respectively. As expected, worse results are associated with low DInSAR coherence between two repeat passes and snow melting periods.

This article is organized as follows. Section II describes data and models including Sentinel 1 SAR data, in situ and auxiliary data as well as snow cover model data and DInSAR forward model. Section III focuses on the retrieval methodology investigating SCA mapping algorithms as well as SH and density inversion techniques. Section IV discusses the results in terms of SCA and height estimates. Section V presents the conclusions, whereas Appendixes give some details on refractive models, the DInSAR sensitivity analysis, winter 2018–2019 conditions, and the density estimation.

II. DATA AND MODELS

SAR data are collected from the Sentinel 1 Copernicus mission, whereas ground data are those available from the area of interest (AOI) in central Italy Apennines. Auxiliary data from a land cover dataset, elevation models, and spaceborne optical radiometers are also used. The SAR interferometric model is based on a plane-wave assumption at the air-snow interface neglecting multiple scattering effects. Details are given in the next paragraphs.

A. AOI and In Situ Data

The AOI is the central Italy Apennines range between 41.5° – 43.5° north latitude and 12.5° – 14.5° east longitude. This region, bounded by the Tyrrhenian Sea on the west side and the Adriatic Sea on the east side, is characterized by a complex orography with deep valleys, medium-short rivers, small lakes, and relatively steep mountains [44]. The highest peak is the Corno Grande at 2912 m above sea level in the Gran Sasso Mountain range which hosts the Calderone glacier, the southernmost glacier (indeed, in the last decades transformed into a *glacieret*) of the European continent.

Within the AOI, there are snow measurement sites, used to validate SAR estimates of snow parameters. These measurements, provided by the Meteomont Avalanche Warning Service (MAWS), are daily acquired within a fenced area (to guarantee the integrity of the snow cover, characterized by requirements about orientation, exposure, and altitude) to allow an institutional snow mantle assessment for reference areas. The measurements provide SH h_s , vertical thermal gradient, and density of the snowpack top layer, which we used to derive the mean snowpack density as described in Appendix D. Fig. 1 shows the location of measurement sites available in central Apennines for this study. The selected measurement sites in the AOI are 32, dislocated quite uniformly within the area itself, typically collecting data during the early morning.

In this work, the period of interest is the winter between December 2018 and February 2019. This winter in Central Italy was characterized by a first small snowfall in mid-December 2018, but the high temperatures during the second part of the



Fig. 1. AOI in Central Italy Apennines. Orange circles indicate snow Meteomont measurement sites, available within the Abruzzo region AOI. Red and yellow rectangles indicate Sentinel 1 subswaths for the descending relative orbit number 22 with acquisition time at 05:11 GMT [using Interferometric Wide (IW) Swath to generate single look complex (SLC)]. The 3 subswaths (from south-east toward north-west) of both coupled master and slave image are shown, whereas the considered intersection area is represented as a white semi-transparent rectangle. Sentinel-1 full swath is 250-km scan with $5 \times 20 \text{ m}^2$ resolution and VV/VH dual polarization.

month led to a complete melt of the snow cover in most of the measurement sites. The largest snowfalls occurred during January 2019, thanks to the formation of a low-pressure system on the Mediterranean Sea. February 2019 was characterized by warm and dry conditions, which caused the retreat of the snow cover and the complete melting of the snow in most of the measurement sites. After February 2019 only snowfall at high elevations, higher than the measurement sites, were observed. For these reasons, we limit our analyses to the period between December 9, 2018 and February 25, 2019. Weather maps that describe the synoptic conditions of winter 2018–2019 are shown in Appendix C.

B. Sentinel-1 SAR and Auxiliary Data

As mentioned, C-band SAR data in SLC format from Sentinel-1A and Sentinel-1B platforms have been used for obtaining differential interferograms. The Sentinel-1 low-Earthorbit (LEO) satellite acquisition modes include:

- 1) Strip Map mode (80-km swath, $5 \times 5 \text{ m}^2$ spatial resolution);
- 2) IW swath mode (250-km swath, $5 \times 20 \text{ m}^2$ spatial resolution);
- 3) Extra-Wide swath mode (400-km swath, $20 \times 40 \text{ m}^2$ spatial resolution);
- 4) Wave mode (20-km swath, $5 \times 5 \text{ m}^2$ spatial resolution).

For the purposes of our study, IW swath products coming from both platforms S1A and S1B are used. The nominal repetitiveness of Sentinel 1 constellation over the AOI is 6 days.

Fig. 1 shows the Sentinel 1 swath, characterized by the relative orbit number 22, acquisition time at 05:11 GMT, a 250 km scan width with $5 \times 20 \text{ m}^2$ resolution and vertical copolar and crosspolar (VV/VH) polarizations. Table I shows the list

 TABLE I

 SENTINEL-1 MASTER/SLAVE COUPLES WITHIN DEC. 9, 2018, AND FEB. 25, 2019. ACQUISITION TIME IS 05:11 GMT (RELATIVE ORBIT NUMBER 22)

Inter ferogram number	Date of acquisition for master	Platform for master	Date of acquisition for slave	Platform for slave
001	2018, Dec. 9	Sentinel 1-A	2018, Dec. 15	Sentinel 1-B
002	2018, Dec. 15	Sentinel 1-B	2018, Dec. 21	Sentinel 1-A
003	2018, Dec. 21	Sentinel 1-A	2018, Dec. 27	Sentinel 1-B
004	2018, Dec. 27	Sentinel 1-B	2019, Jan. 2	Sentinel 1-A
005	2019, Jan. 2	Sentinel 1-A	2019, Jan. 8	Sentinel 1-B
006	2019, Jan. 8	Sentinel 1-B	2019, Jan. 14	Sentinel 1-A
007	2019, Jan. 14	Sentinel 1-A	2019, Jan. 20	Sentinel 1-B
008	2019, Jan. 20	Sentinel 1-B	2019, Jan. 26	Sentinel 1-A
009	2019, Jan. 26	Sentinel 1-A	2019, Feb. 1	Sentinel 1-B
010	2019, Feb. 1	Sentinel 1-B	2019, Feb. 7	Sentinel 1-A
011	2019, Feb. 7	Sentinel 1-A	2019, Feb. 13	Sentinel 1-B
012	2019, Feb. 13	Sentinel 1-B	2019, Feb. 19	Sentinel 1-A
013	2019, Feb. 19	Sentinel 1-A	2019, Feb. 25	Sentinel 1-B

of Sentinel-1 acquisition dates for the SLC-IW images. This set is used to generate 13 interferograms within the period of interest. For all generated interferograms, the master and slave image coupling is based on adjacent dates. Both unwrapped interferometric phase maps and coherence maps are obtained by a standard process of filtering and unwrapping, as better explained later. For each Sentinel-1 SLC-IW image, the backscattering coefficient in both vertical copolar and crosspolar polarizations is also derived.

The applicability of DInSAR technique to the snow cover parameter retrieval is strongly limited by the relatively poor repetitiveness of the Sentinel-1 constellation, which is indeed quite high if compared to other satellite SAR missions. Moreover, the DInSAR signal features shows ambiguities over complex orographic terrains and in presence of variable atmospheric conditions between master/slave acquisitions. To overcome as much as possible these SAR observation limits, auxiliary information and data are employed.

Table II shows the list of external auxiliary data, used in combination with Sentinel-1 data. The table reports the following.

- 1) DEM, used to enhance the accuracy of the interferometric data processing and retrieval algorithms [5].
- Atmospheric correction model data, used to mitigate Din-SAR interferogram distortions caused by water vapor variable pattern between master and slave image dates [53].
- A land cover map, used to assist with dry and wet snow classification [54].
- Snow-cover products obtained from the Terra-satellite's "Moderate Resolution Imaging Spectroradiometer" (MODIS) instrument, used to assist in the discrimination of snow-covered areas from snow-free areas [14].

The simulation of fundamental snow cover properties is based on the Alpine3D model simulations. Alpine3D is a Lagrangian

TABLE II Auxiliary Data, Used in This Work, With Its Sources and Main Characteristics

Data	Source	Explanation
Terrain Elevation Map	Shuttle Radar Topographic Mission (3s)	DEM Sampling resolution: 3 s (~90 m) Resampling method: Bicubic, Interpolation Resampling type: Bisinc 21 Point Interpolation
Land Cover maps	Corine Land Cover (CLC)	Database of Land Cover type derived from Sentinel-2 and Landsat-8 data. Provided by Copernicus
Snow Cover Fraction (SFC) satellite-based map	MODIS	Cloudless SCF at 500 m horizontal resolution. For each cloud pixel, we made a linear interpolation between previous and consecutive acquisitions of not cloud-covered pixels, and we inferred the SCF for cloud pixels.
Atmospheric water-vapor correction map	GACOS	Absolute phase correction for C-Band due to water vapor derived from an atmospheric water-vapor correction model using Global Positioning System (GPS) data.
Maps of snow cover parameters	Alpine3D	Snow cover Lagrangian dynamic model forced with forecasted weather data at 3 km spatial resolution and 1-h temporal resolution. Selected outputs: snow depth, SD, snow water fraction, snow temperature.

TABLE III Definitions, Acronyms, and Symbols

Variables (units)	Symbols
SH (m)	h_s
Snow relative height (m)	Δh_s
SWE (m)	h _{swe}
Snow complex relative permittivity (adim)	$\varepsilon_s = \varepsilon_{Rs} - j\varepsilon_{Js}$
SCA (m ²)	A_s
Dry snow complex relative permittivity (adim)	$\varepsilon_{ds} = \varepsilon_{Rds} - j\varepsilon_{Jds}$
Snow liquid water volumetric fraction (adim)	f_w
Snow density (SD) (g/cm ³)	$ ho_s$
Dry SD (g/cm ³)	$ ho_{ds}$
Water density (g/cm ³)	$ ho_w$
Snow interferometric phase (°) at pp polarization	$\Delta \Phi_{pps}$
Interferometric coherence (adim)	γ_{pp}
SAR incident wave number (adim)	k
SAR incidence angle (°)	θ
SAR local incidence angle (°)	θ_l
SAR range (m)	R
Topography slope angle (°)	θ_t

numerical model used to simulate high-resolution surface processes, and in particular snow processes in mountainous regions [44], [47]. The Alpine3D model is driven by external data such as measurements at stations or outputs from meteorological models. The Alpine3D model consists of several component modules.

- 1) The SNOWPACK one-dimensional (1-D) model of vegetation, snow, and soil [47].
- A 3-D radiation balance model (which considers shortwave scattering and longwave emission from terrain and tall vegetation).
- 3) A 3-D drifting snow model solving a diffusion equation for suspended snow and a saltation transport equation. Examples of the use of the Alpine3D model include the analysis of snow cover dynamics for avalanche warning, studies on permafrost development and vegetation changes for the monitoring of climate change evolution, as well as the estimation of soil moisture aimed at obtaining highly accurate meteorological and flood forecasting models.

The Alpine3D model, driven by forecasted weather data, has been validated in the AOI [44]. Results showed that Alpine3D is able to reproduce the observed SH and SWE with a bias of 6 cm and 35 mm respectively, with a correlation coefficient of 0.87 for SH and 0.74 for SWE.

The Alpine3D map dataset has been generated with a 3-km spatial resolution at 1-h temporal resolution and has been resampled to a reference resolution of 100 m, which has been adopted in this study to ensure uniformity across all employed spatial data. We deem such a resolution to be sufficient to characterize snow properties for a relatively large AOI, and to be an optimal tradeoff in terms of computational costs [1], [3]. For each pixel of the AOI, the Alpine3D model provides altitude, SH, SD, and snow liquid water fraction as well as the temperature profile. In this work, the generated Alpine3D dataset is basically used to train snow estimation models from DInSAR data.



Fig. 2. Geometry of DInSAR signal propagation in a snow deposit, characterized by a single uniform snow layer. The yellow ray path refers to the snow absence, whereas the red one to the snow presence inducing refraction. Planar interfaces are assumed, whereas the ground plane is tilted due to the ground local slope where θ is the incidence angle, θ_l is the local incidence angle, θ_t is the terrain slope angle, and θ_s is the Snell refraction angle, Δh_s is the relative SH, Δz_s is the relative snow thickness. Other symbols are defined in the main text, apart from **g** that represents the Earth gravitational acceleration vector.

C. SAR Differential-Phase Forward Model

The snow estimation methodology from SAR differential interferometry, based on satellite repeat passes above the same scene, takes advantage of the nearly linear dependence existing between the SAR signal path delay and SH h_s [7], [11]. Even though widely explained in literature, we briefly summarize here the principles to introduce the basic notation.

Table III lists the main definitions, acronyms, and symbols, used throughout the text.

Fig. 2 illustrates how the SAR signal, supposed to be locally approximated by a plane wavefront, is affected by the snow refraction causing a path delay with respect to the same surface without snow. Note that the incidence angle θ is the angle between the normal of the flat earth (ellipsoid) and the satellite pointing, whereas the local incidence angle θ_l is the angle between the normal of the topography and the satellite pointing. Depending on the relief within the scene and the topography slope angle θ_t , these two angles can be significantly different (if $\theta_t = 0$ in Fig. 2 then $\theta_l = \theta$). Each interface in Fig. 2 is considered as a specular plane, for simplicity, and possible multiple scattering effects within the snow layer (due to ice grains, ice lenses, and/or vertical refrozen structures larger or comparable with SAR central wavelength) are disregarded [27], [29].

In Fig. 2, we exemplify the problem geometry for one planeparallel uniform non-absorbing isotropic snow layer above a surface with a local slope [25], [29]. The snow complex permittivity can be expressed by [13]

$$\varepsilon_s\left(\rho_{ds}, f_w\right) = \varepsilon_{Rs} - j\varepsilon_{Js} = \varepsilon_{rs} \,\varepsilon_0 = \left(\varepsilon_{Rrs} - j\varepsilon_{Jrs}\right)\varepsilon_0 \tag{1}$$

with ε_{Rs} and ε_{Js} the real and imaginary part of ε_s (with *j* the imaginary unit), ε_{rs} the relative snow permittivity with ε_{Rrs} and ε_{Jrs} the real and imaginary part respectively, and ε_0 the vacuum permittivity. The adopted snow permittivity models, mainly depending on the dry SD ρ_{ds} and liquid water fraction f_w as highlighted in (1) to include wet snow, are reported for completeness in Appendix A [55], [57].

Note that the SD ρ_s differs from the dry SD ρ_{ds} due to the presence of liquid water fraction. Moreover, snowflakes are usually compressed by weight after ground deposition (snow settling) and some days later, recrystallization may form ice grains that are spherical at first, vertically extended at later phases, and which typically increase in size over time (snow metamorphism) [55], [56]. This snow microstructures typically cause macroscopic anisotropy that reflects into polarization-dependent effective dielectric constant of the air-ice mixture, perturbed by liquid water from the melting process. As a result, the snow complex permittivity in (1) can be polarization dependent and snow can be modeled as an air medium populated by aligned spheroidal ice particles using the Maxwell-Garnet theory [26].

The point P on the ground is covered by a snow layer whose relative height is (see Fig. 2)

$$\Delta h_s = h_s \left(t_2 \right) - h_{s0} \left(t_1 \right) = \Delta z_s / \cos \theta_t \tag{2}$$

with h_s the SH at time t_2 (> t_1) and h_{s0} the reference surface (ground surface at time t_1) height, whereas the quantity Δz_s in (2) is the relative snow thickness, measured along the terrain normal. In (2), θ_t represents an important feature in complex orography. If R_s is the distance between the SAR sensor and snow surface, the plane wave fronts of the signal will be refracted at point Q and, covering the distance ΔR_s , will reach the point P.

In absence of snow, no refraction would occur and propagation distance in air to point P is $R_a + \Delta R_a$. If $n_s = \sqrt{\varepsilon_{Rrs}}$ is the snow refractive index real part, the geometric path difference ΔR between the two conditions (i.e., snow presence and absence) is determined by

$$\Delta R = (R_s + n_s \,\Delta R_s) - (R_a + \Delta R_a) \tag{3}$$

being $\Delta R_s = \Delta z_s / \cos \theta_s$ and $\Delta R_a = \Delta R_s \cos(\theta_l - \theta_s)$. By applying trigonometry and using the Snell law $\sin \theta = n_s \sin \theta_s$ [13], we obtain

$$\Delta R = -\Delta z_s \left(\cos \theta_l + \frac{\sin \theta_l \sin \theta_s - n_s}{\cos \theta_s} \right)$$
$$= -\Delta z_s \left(\cos \theta_l - \sqrt{\varepsilon_{Rrs} - \sin^2 \theta_l} \right). \tag{4}$$

Changes in snow layer thickness Δz_s by a positive (due to snowfall) or negative (due to snow ablation) amount reflect into a change of the SAR wave front two-way differential phase at pp co-polarization given by $\Delta \Phi_{pps} = 2 (2\pi \Delta R/\lambda)$. Using (2), it holds for the differential phase due to snow

$$\Delta \Phi_{pps} = f \frac{4\pi \Delta R}{v_s} = -2k \frac{\left(\cos \theta_l - \sqrt{\varepsilon_{Rrs} - sin^2 \theta_l}\right)}{\cos \theta_t} \Delta h_s$$
⁽⁵⁾

being $k = 2\pi/\lambda$ with *f* the SAR operating frequency and $\lambda = v_s/f_0$ the SAR wavelength in the snow layer propagating at phase velocity $v_s = c/n_s$ with *c* the vacuum light velocity. For Sentinel-1 DInSAR, p = v using the copolar returns at vertical (*v*) polarization. Note that for an anisotropic snow layer, due to its polarization-dependent effective permittivity, $\Delta \Phi_{vvs}$ is not equal to $\Delta \Phi_{hhs}$ thus causing a copolar differential phase shift (or phase difference) $\Delta \Phi_{cos} = \Delta \Phi_{vvs} - \Delta \Phi_{hhs}$, fairly well correlated at X band to the h_s [26]. In the following text, we will keep the notation $\Delta \Phi_{pps}$ for generality.

The previous relation (5) constitutes the basic forward model that can be used to translate the dry or wet snow thickness into DInSAR differential phase $\Delta \Phi_s$. For a dry snow uniform isotropic layer, using the permittivity model (A.1) illustrated in Appendix A, it can be shown that (3) can be approximated with an error less than 3% (for θ less than 50°) by [27]

$$\Delta \Phi_{pps} \cong ka_c \frac{\left(b_c + \theta_l^{5/2}\right)}{\cos \theta_t} \frac{\rho_s}{\rho_w} \Delta h_s = ka_c \frac{\left(b_c + \theta_l^{5/2}\right)}{\cos \theta_t} \Delta h_{swe}$$
(6)

being the terms $a_c = 1$ (average optimal value) and $b_c = 1.59$, whereas the relative SWE height Δh_{swe} is defined as follows:

$$\Delta h_{swe} = \frac{\Delta z_{swe}}{\cos \theta_t} = \frac{1}{\cos \theta_t \rho_w} \int_{z_0}^{z_s} \rho_s (z) dz = \frac{\rho_s}{\rho_w} \Delta h_s$$
(7)

with Δz_{swe} the relative SWE thickness (see Fig. 2). The last term in (7) holds only for a uniform snow layer.

Moreover, the approximate expression in (6), in terms of Δh_{swe} , holds also for a non-uniform snow layer with a variable density below the air-snow interface, i.e., [27]

$$\Delta \Phi_{pps} = ka_c \frac{\left(b_c + \theta_l^{5/2}\right)}{\cos \theta_t} \sum_{i=1}^N \frac{\rho_s(i)}{\rho_w} \Delta h_s(i)$$
$$= ka_c \frac{\left(b_c + \theta_l^{5/2}\right)}{\cos \theta_t} \Delta h_{swe} \tag{8}$$

where N is the number of the different snow *i*th layers.

For a wet snow layer, $\Delta \Phi_s$ becomes a function of the snow water fraction so that (6) must be replaced with the more general expression (5) [29], [40]. As an example, since n_s is a function



Fig. 3. DInSAR differential phase $\Delta \Phi_{pps}$ versus snow relative height Δh_s for some values of liquid water fraction and dry SD, using the analytical forward model in (5) with $\theta_t = 0^\circ$ and Wiesmann-Mätzler snow permittivity model of Appendix A. (Top) $\theta = 20^\circ$. (Bottom) $\theta = 40^\circ$.

of ρ_{ds} and f_w , using the Wiesmann-Mätzler snow permittivity model in Appendix A, Fig. 3 shows how $\Delta \Phi_s$ is affected by different values of ρ_{ds} and f_w for two incidence angles $\theta = 20^{\circ}$ and $\theta = 40^{\circ}$, assuming $\theta_t = 0$ and setting the frequency at 5.6 GHz, the one of Sentinel-1 SAR. $\Delta \Phi_{pps}$ is less sensitive to h_s variation for low snow densities and lower incidence angles, even though the impact of the latter is relatively low. The increase in liquid water fraction enhances the sensitivity in all cases. For Δh_s larger than a few centimeters, we expect a SAR phase folding issue at C band (that is, $\Delta \Phi_{pps} > 180^{\circ}$).

The analytical dependence between $\Delta \Phi_{pps}$ and Δh_s in (5) can also be used for carrying out an uncertainty analysis to evaluate how errors on θ_l , ρ_{ds} , and f_w would influence $\Delta \Phi_{pps}$. Using the first-order error propagation theory and assuming the statistical independence of all uncertainties, the DInSAR differential phase standard deviation $\sigma_{\Delta \Phi_{pps}}$ can be expressed by the square root of the error variance sum weighted by the respective square partial derivative

$$\sigma_{\Delta\Phi_{pps}} = \sqrt{\left(\frac{\partial\Delta\Phi_{pps}}{\partial\theta_l}\right)^2 \sigma_{\theta_l}^2} + \left(\frac{\partial\Delta\Phi_{pps}}{\partial\rho_{ds}}\right)^2 \sigma_{\rho_{ds}}^2 + \left(\frac{\partial\Delta\Phi_{pps}}{\partial f_w}\right)^2 \sigma_{f_w}^2 \tag{9}$$

being σ_{θ_l} , $\sigma_{\rho_{ds}}$, and σ_{f_w} the standard deviations of θ_l , ρ_{ds} , and f_w , respectively. The three partial-derivative terms of (9) are analytically derived and discussed in Appendix B using, as

an upper bound, the linear-mixing model for the snow relative permittivity described in Appendix A.

III. METHODOLOGY

The previous section was dedicated to the description of data and DInSAR forward models. Starting from that analysis, this section will be devoted to the illustration of the proposed inversion approaches that will be applied in the next sections.

A. Retrieval Methodology

The adopted processing methods include DInSAR standard processing schemes, inversion methods to obtain snow properties estimates from SAR sensor measurements, decision tree, and validation steps. Three main branches are visible in Fig. 4.

- 1) SAR and DinSAR data processing modules.
- 2) SCA classification modules.
- 3) Snow cover parameter estimation modules.

For the retrieval processes, we have opted for two separate ANNs as inversion methods [60]: one for the snow classification and one for the SH and density estimation. Indeed, the diagram of Fig. 4 can be easily adapted, with minor modifications, to represent other retrieval methods, for instance, based on analytical and statistical approaches as discussed later. The overall ANN-based retrieval scheme produces for the whole area.

- 1) Dry/wet SCA estimated maps.
- 2) SH estimated maps.
- SD or SWE estimated maps. Each step is discussed in what follows.

To properly handle data fusion, Fig. 4 shows the SAR-based integrated approach for the ANN retrieval of SCA, type, height, and density, coupling Alpine3D model and auxiliary data. Input sources include the following.

- 1) Data from satellite SAR.
- Data from auxiliary remote sensing observations such as spaceborne optical and near-infrared sensors (these data are used for the training of the ANN, which performs the snow cover classification and also to validate results).
- Data products derived from satellite data such as DEMs and land cover classification maps (these data are used within the ANN to improve the precision of the results).
- 4) Data from numerical models for snow cover evolution simulations (these data are used for the physical-analytical snow model, which requires the knowledge of some data not retrievable with satellites like SD and liquid water fraction).
- 5) Ground-based data (e.g., in situ measurements: these data are used to validate results). As mentioned, all data listed in Tables II and III have been reprojected and bilinearly interpolated onto a common resample grid with a linear resolution of about 100 m [1], [3].

As discussed in [27], the quality of the measured interferometric coherence γ_{ppsm} between two consecutive acquisitions is crucial for the DInSAR method application. Incoherent gaps between interferograms tend to degrade the overall DInSAR signal-to-noise ratio. The possible coherence loss on snowcovered areas may be caused by the variation of the liquid water



Fig. 4. DInSAR-based integrated approach for retrieving snow cover properties. Symbols are defined in the main text as well as processing modules (indicated by rectangular boxes), output products (indicated by elliptic shapes), and auxiliary inputs (indicated by rounded rectangles).

fraction due to positive temperatures, melting processes, and/or rainfall. Coherence loss may also occur when strong winds lead to a snow erosion and redistribution. Another possible cause of the coherence loss is the snow structure collapse by gravity or external forces over weak or thin snow layers. Finally, the vegetation cover above the snow cover can also degrade the DIn-SAR coherence due to the snow weight on the plant branches. Using DInSAR lower frequencies, the coherence temporal loss is slower, the penetration depth is larger due to less impact of liquid water fraction and phase-wrapping errors are usually smaller, but SH and density retrieval is less accurate due to longer wavelengths. In this respect, the use of C-band SAR data may be a good compromise.

In DInSAR, the measured differential phase $\Delta \Phi_{ppsm}$ at snow surface is computed from the measured absolute phase Φ_{ppsm} of 2 repeated passes at time t_1 (master image) and t_2 (slave image), that is [14]

$$\Delta\Phi_{ppsm}(x,y) = \Phi_{ppsm}(x,y,t_2) - \Phi_{ppsm}(x,y,t_1) \quad (10)$$

where Φ_{ppsm} is the two-way propagation delay due to the radar wave refraction in the snow cover, referred to snow-free conditions (see Fig. 2).

In the flow diagram of Fig. 4, master and slave SAR SLC measurements are processed according to a consolidated scheme. Assuming a sufficiently high coherence between two SAR acquisitions over the same snow-covered scene, the measured snow interferometric phase $\Delta \Phi_{ppsm}$ (with the subscript *m* standing for *measured* in the following text) at *pp* polarization is obtained for the (*x*,*y*) pixel from the measured interferometric phase $\Delta \Phi_{ppm}$ by (e.g., [12], [13]):

$$\Delta \Phi_{ppsm} (x, y) = \Delta \Phi_{ppm} (x, y) - \Delta \Phi_{ppflat} (x, y) - \Delta \Phi_{pptopo} (x, y) - \Delta \Phi_{ppatm} (x, y) - \Delta \Phi_{ppnoise} (x, y)$$
(11)

where $\Delta \Phi_{ppflat}$ and $\Delta \Phi_{pptopo}$ are the differential phase differences due to changes of the relative distance between satellite

and target for flat Earth and for topography, respectively, whereas $\Delta \Phi_{ppatm}$ is generated from changes in atmospheric propagation due to ionosphere and troposphere, and $\Delta \Phi_{ppnoise}$ is differential phase noise.

Sentinel-1 C-band DInSAR data processing, underlined in (10), includes the following.

- 1) TOPSAR splitting.
- 2) Orbit file application and geocoding.
- 3) Enhanced spectral diversity and interferogram generation.
- 4) TOPSAR debursting.
- 5) Topographic phase removal.
- 6) Goldstein phase and multi-look filter.
- 7) Masking and phase unwrapping.
- 8) Terrain correction.

The phase unwrapping is based on the Statistical-Cost, Network-Flow Algorithm for Phase Unwrapping algorithm, which is an effective and efficient two-dimensional phase unwrapping algorithm [63]. The atmospheric phase screen correction, derived from the Generic Atmospheric Correction Online Service for InSAR (GACOS) products, has been applied to the unwrapped phase values [53]. The SAR local incidence angle information, required in (5), is obtained as a subproduct of the interferogram terrain correction process. The final output contains Sentinel-1 unwrapped phase interferograms, coherence, and local incidence angle maps. SAR SLC data are also processed to obtain the SAR backscattering copolar coefficients.

B. SCA Classification

For the SCA mapping, we need to address two separate objectives.

- 1) Identifying the presence or absence of snow (primary binary mask).
- Discriminating the wet snow from dry snow within the SCA (secondary binary mask).

When using only SAR backscattering data, the Nagler-Rott empirical approach can be adopted [10], [28]. This technique

uses the SAR backscattering coefficient σ^0 to discriminate between wet and dry snow. For each pixel (*x*,*y*), two backscattering measurements are considered: 1) $\sigma^0_{vvmref}(x,y)$, the copolar vertically polarized data from a reference SAR image; 2) $\sigma^0_{vvm}(x,y)$, the copolar vertically polarized data from the SAR image during the snow cover period.

According to the Nagler-Rott method, the copolar ratio $r_{vv}(x,y)$ indicates the presence of wet snow in case its value is below a threshold, and the presence of dry snow otherwise [10]

$$r_{vv}(x,y) = \frac{\sigma_{vvm}^{0}(x,y)}{\sigma_{vvmref}^{0}(x,y)} = \begin{cases} \leq r_{0wet} \text{ Wet snow} \\ > r_{0wet} \text{ Dry snow} \end{cases}$$
(12)

where r_{0wet} is the wet-snow threshold (in dB), very often empirically set to about -3 dB. A further extension of (12) is considering also the cross-polar backscattering coefficient $\sigma_{vvm}^0(x, y)$ and θ_l so that the combined ratio r_c is given by [28]

$$r_{c}(x, y, \theta_{l}) = W(\theta_{l}) \frac{\sigma_{vh}^{0}(x, y)}{\sigma_{vhref}^{0}(x, y)} + [1 - W(\theta_{l})] r_{vv}(x, y)$$
(13)

where W is an empirically derived angular weighting tree function [28]. To identify the wet snow cover, the ratio $r_c(x, y, \theta_l)$ is typically set below the threshold of -2 dB.

A limitation of the Nagler-Rott approach is the exploitation of SAR backscattering data only. From Sentinel-1 DInSAR acquisitions, we can also extract the *vv*-copolar differential phase difference $\Delta \Phi_{vvsm}$ and coherence γ_{vvsm} as well as integrate auxiliary data available from Table II. To perform an effective data fusion, we have designed an ANN and combined it with a fuzzy logic algorithm to obtain snow/no-snow and dry-snow/wet-snow cover maps.

The proposed Snow Cover Classification ANN (SCC-ANN) employs a feed-forward backpropagation architecture with a Bayesian regularization. The SCC-ANN is composed by a single hidden layer with 14 neurons using a sigmoid activation function. Formally speaking, we can write for each pixel (x, y)

$$\hat{\mathbf{x}}_{scc} (x, y) = \begin{bmatrix} \hat{f}_{SC} (x, y) \\ \hat{f}_{w} (x, y) \end{bmatrix}$$
$$= ANN_{SCC} \left\{ \boldsymbol{y}_{mSAR} (x, y) ; \boldsymbol{y}_{ap} (x, y) \right\} \quad (14)$$

where $\hat{\mathbf{x}}_{scc}(x, y)$ is the estimated snow cover classification parameter vector, whereas $\boldsymbol{y}_{mSAR}(x, y)$ and $\boldsymbol{y}_{ap}(x, y)$ are the input data parameter vectors extracted for each pixel from Sentinel-1 SAR observations and a priori auxiliary data in Table II, respectively. Specifically,

1) The input measurement vector $\boldsymbol{y}_{mSAR}(x, y)$ includes: i) DinSAR coherence $\gamma_{vvsm}(x, y)$; ii) SAR copolar backscattering coefficient $\sigma^0_{vvm}(x, y)$ for master and slave images; iii) SAR backscattering cross-polar coefficient $\sigma^0_{vhm}(x, y)$ for master and slave images; iv) SAR backscattering copolar coefficient $\sigma^0_{vvmref}(x, y)$ at the reference (nosnow) date; v) SAR backscattering cross-polar coefficient $\sigma^0_{vhmref}(x, y)$ at the reference (no-snow) date.

- 2) The input vector $\boldsymbol{y}_{ap}(x, y)$ includes: i) local incidence angle $\theta_l(x, y)$; ii) SRTM DEM data $h_{topo}(x, y)$; iii) CORINE land cover labels $L_c(x, y)$.
- The output vector \$\hfrac{x}_{scc}(x, y)\$ includes two snow parameters: i) the fractional snow cover (FSC) \$f_{SC}(x,y)\$, that is the percentage of snow cover for each pixel; ii) the volumetric liquid water fraction \$f_w(x,y)\$, that is the percentage of liquid water in a unit volume of snow.

To train SCC-ANN, the output variables $f_{SC}(x,y)$ and $f_w(x,y)$ have been extracted from MODIS and Alpine3D respectively, after a bilinear interpolation and resampling onto a common grid at 100-m resolution. By selecting more than 10.000 samples within the area/period of interest and dividing the whole set into a validation (15%) and test (15%), the SCC-ANN shows relatively good results, as discussed later.

Based on the SCC-ANN estimated $\hat{f}_{SC}(x, y)$, we can perform the first task to identify the presence/absence of snow (primary binary mask). This can be accomplished by a simple FSC threshold algorithm, like the one of Nagler-Rott in (12), that is

$$\hat{f}_{SC}(x,y) = \begin{cases} \leq f_{0snow} \text{ Snow absence} \\ > f_{0snow} \text{ Snow presence} \end{cases}$$
(15)

where f_{0snow} is an arbitrary threshold (in %), set to an optimal value during the training stage (typically between 5% and 25%). The SCA is finally computed by summing up all the snow pixels satisfying snow presence in (15).

We can perform the second task to discriminate the dry/wet snow (secondary binary mask) using the SCC-ANN estimated $f_{SC}(x,y)$ and $\hat{f}_w(x,y)$. Instead of relying on a fixed threshold for the fraction of water, we have adopted a fuzzy logic approach also involving the snow surface temperature $T_s(x,y)$ as a third parameter coming from the Alpine3D simulations. Fuzzy logic is a form of many-valued logic in which the truth value of variables may be any real number between 0 and 1, extending *de facto* the Boolean logic [62]. Fuzzy logic models or sets are an extremely convenient and suitable form of representing uncertain information. Two membership functions M_w and M_t are defined for the snow water fraction and snow surface temperature, respectively:

$$\begin{cases} M_w \left[\hat{f}_w \left(x, y \right) \right] = Ramp \left[\hat{f}_w \left(x, y \right), w_1, w_2, \right] \\ M_t \left[T_s \left(x, y \right) \right] = Ramp \left[T_s \left(x, y \right), t_1, t_2, \right] \end{cases}$$
(16)

where in our case the coefficients are $w_1 = 0.1\%$, $w_2 = 0.8\%$, $t_1 = -0.2$ °C, $t_2 = -0.1$ °C, being *Ramp* the ramp function

Ramp
$$(x, a, b) = \begin{cases} 0, \ x \le a \\ x (b-a) - a/(b-a), \ a < x \le b \\ 1, \ x > b \end{cases}$$
 (17)

If the inference function *I* is defined through a multiplicative formula by means of

$$I\left[\hat{f}_{w}\left(x,y\right),T_{s}\left(x,y\right)\right] = M_{c}\left[\hat{f}_{w}\left(x,y\right)\right]M_{t}\left[T_{s}\left(x,y\right)\right]$$
(18)

then the following defuzzification rule is applied to the inference function:

$$I\left[\hat{f}_{w}\left(x,y\right),T_{s}\left(x,y\right)\right] = \begin{cases}>I_{0} \text{ Wet snow}\\\leq I_{0} \text{ Dry snow}\end{cases}$$
(19)

where the threshold I_0 is usually set to 0.5, equal to a uniform probability of the 2 fuzzy states.

C. SH and Density Estimation

Once the SCA and wet/snow mask is obtained, we can approach the estimation of $\Delta h_s(x, y)$ for each image pixel from SAR and DInSAR observations. In this respect, satellite optical data and products are not useful due to the high snow and cloud extinction at those wavelengths. Microwave remote observations are the only ones that can provide a physically based estimate of the snow depth.

The artificial Neural Network Inversion (NNI) method is based on the use of an ANN, as already done in Section III-B. The snow depth estimation operation is performed by a dedicated neural network, named SnowPack Estimation ANN (SPE-ANN). The architecture of the SPE-ANN is like the SCC-ANN one, using a Bayesian regularization with 1 hidden layer and 14 neurons.

Formally speaking, we can write for each pixel (x,y)

$$\hat{\mathbf{x}}_{spe} (x, y) = \begin{bmatrix} \Delta \hat{h}_s (x, y) \\ \hat{\rho}_s (x, y) \end{bmatrix}$$
$$= ANN_{SPE} \{ \boldsymbol{z}_{mSAR} (x, y) ; \boldsymbol{z}_{ap} (x, y) \} (20)$$

where $\hat{\mathbf{x}}_{spe}(x, y)$ is the estimated snow cover parameter vector, whereas $\mathbf{z}_{mSAR}(x, y)$ and $\mathbf{z}_{ap}(x, y)$ are the input data parameter vectors extracted for each pixel from SAR observations and a priori integrative data in Table II, respectively. Specifically, for Sentinel-1 SAR data

- 1) input vector $\boldsymbol{z}_{mSAR}(x, y)$ includes: i) DinSAR coherence $\gamma_{vvs}(x,y)$; ii) DInSAR co-polar unwrapped differential phase $\Delta \Phi_{vvs}(x, y)$; iii) DInSAR cross-polar unwrapped differential phase $\Delta \Phi_{vhs}(x, y)$.
- 2) input vector $z_{ap}(x, y)$ includes: i) local incidence angle $\theta_l(x,y)$; ii) atmospheric water vapor correction maps $\Delta \Phi_{atm}(x, y)$, derived from the GACOS algorithm; iii) SRTM DEM data $h_{topo}(x,y)$; iv) CORINE land cover labels $L_c(x,y)$.
- 3) output vector $\hat{\mathbf{x}}_{spe}(x, y)$ includes two snow parameters: i) snow relative height $\Delta \hat{h}_s(x, y)$; ii) SD $\hat{\rho}_s(x, y)$.

To train SPE-ANN, the output variables $\Delta \hat{h}_s(x, y)$ and $\hat{\rho}_{ds}(x, y)$ have been extracted from Alpine3D, after a bilinear interpolation and resampling onto a common grid at 100-m resolution.

To intercompare NNI with other estimation approaches, we have implemented the following inversion algorithms:

- 1) Physical-Analytical Inversion (PAI);
- MonteCarlo-based Linear Regressive Inversion (LRI-MC);
- 3) Alpine3D-based Linear Regressive Inversion (LRI-A3);
- 4) MonteCarlo-based Max Likelihood Inversion (MLI-MC);
- 5) Alpine3D-based Max Likelihood Inversion (MLI-A3).

A brief description of the these inversion algorithms is given hereafter.

1) The *PAI algorithm* is obtained by analytically inverting the DInSAR equation in (5), that is:

$$\Delta \hat{h}_{s}(x,y) = \frac{1}{\Delta \Phi_{ppsm}(x,y)} \frac{\cos \theta_{t}}{2k \left(\sqrt{\varepsilon_{Rrs} - \sin^{2} \theta_{l}} - \cos \theta_{l}\right)}.$$
(21)

The PAI inversion algorithm requires that θ_l and θ_t are known, but more importantly an assumption on ρ_{ds} and f_w to estimate ε_{Rrs} in (21) (see Appendix A). The latter parameters are not usually known, and their arbitrary assignment constitutes the main limitation of this inversion method.

2) The *LRI algorithms* are based on the statistical linearregression approach, described in the following equation:

$$\Delta h_s(x,y) = a_{h0} + a_{h1} \Delta \Phi_{ppsm}(x,y) \qquad (22)$$

where the regression coefficients a_{h0} and a_{h1} are derived from the DInSAR forward model simulations.

The latter can be based on: 1) a MonteCarlo random approach, where the input model parameters (i.e., h_s , ρ_{ds} , f_w in (5) using (A.2)) are uniformly varied between a minimum and a maximum value (the algorithm is then named **LRI-MC**); 2) Alpine3D numerical outputs, where the same input model parameters are provided by the Alpine3D simulations described in Section II-D (the algorithm is then named **LRI-A3**).

In the LRI-MC, we have set these ranges: $0.05 \le \Delta h_s \le 1.00$ m, $50 \le \rho_{ds} \le 500$ kg/m³, $0 \le f_w \le 4\%$. The main difference between LRI-MC and LRI-A3 is that the input snow parameters are randomly selected in the first case and physically constrained in the second case. This implies that there are Monte Carlo triplets that do not have necessarily a physical meaning.

3) The *MLI algorithms* are statistical approaches where the error probability density function is maximized, that is under a Gaussian hypothesis its negative argument is minimized [61]. If a forward model is available, minimizing the error (or objective) function implies to minimize the square difference between the forward model observable $\Delta \Phi_{pps}$ [derived from (5)] and the corresponding measured one $\Delta \Phi_{ppsm}$. Formally speaking, using the DInSAR observable $\Delta \Phi_{pps}(x, y)$, we can write the estimation through the objective function as

$$\begin{bmatrix} \Delta \hat{h}_{s}(x,y) \\ \hat{\rho}_{s}(x,y) \end{bmatrix} = argmin \left\{ \left[\Delta \Phi_{ppsm}(x,y) - \Delta \Phi_{pps}(x,y) \right]^{2} \right\}$$
(23)

where *argmin* is argument-of-the-minimum operator restituting the elements $\Delta \hat{h}_s$ and $\hat{\rho}_{ds}$ of the domain at which the function values are minimized. With respect to PAI



Fig. 5. Example of relative SH Δh_s map, simulated by Alpine3D from Dec. 27, 2018 till Jan. 2, 2019.

and LRI algorithms, the MLI technique can provide a statistical estimate of both SH and density.

As for LRI, we can distinguish again between a Monte Carlo random and an Alpine3D model approach, defining **MLI-MC** and **MLI-A3** respectively. For computational purposes, the minimization can be carried out by searching in a pregenerated database and setting a confidence interval to provide an associate uncertainty.

For h_s estimate, we can further investigate a method combining MLI and Alpine3D simulations. This combined approach, called **MLI-A3C**, can use Alpine3D model values of SH whenever the DInSAR coherence ρ_{vvs} is below a given threshold (e.g., 0.3). This approach should improve the accuracy of the estimated $\Delta \hat{h}_s$ when the DinSAR coherence is too low to provide reliable estimates of SH. The combination between MLI estimates and Alpine3D model values can be linearly weighted by the coherence γ_{vvm} as follows:

$$\Delta \hat{h}_s (x, y) = \gamma_{vvm} \Delta \hat{h}_s^{MLI}(x, y) + (1 - \gamma_{vvm}) \Delta \hat{h}_s^{A3D}(x, y)$$
(24)

where $\Delta \hat{h}_s^{MLI}$ and $\Delta \hat{h}_s^{A3D}$ are the snow relative heights from MLI and Alpine3D model, respectively.

IV. RESULTS

After introducing the retrieval methodologies, in this section, we will illustrate the numerical results obtained from the forward and inverse models.

A. Snow Cover Model Simulations

An example of a SH variation map, simulated from the Alpine3D model between Dec. 27, 2018 and Jan. 2, 2019, is shown in Fig. 5. The reason for the snow depletion in many high-altitude areas is due to the lack of snowfall and temperature raising during the considered week. This h_s variation map is a

useful representation when compared with DInSAR interferograms.

Fig. 6 shows bi-dimensional histogram of dry SD and volumetric liquid water fraction, generated by the Alpine3D snow cover model within the area (see Fig. 1) and period of interest. SD values range from 50 till 400 kg/m³, whereas f_w spans from 0% up to 6% denoting relatively dry samples, probably due to the exclusion of the last winter months. Note that a minimum threshold of h_s equal to 1 cm is applied to plot data in Fig. 6 (smaller values are considered not numerically significant).

The DInSAR forward model in (5) can be coupled with the numerical outputs of the Alpine3D snow cover model for the area and period of interest. Fig. 7 shows the simulated $\Delta \Phi_{pps}$ values using the Alpine3D dataset of Δh_s , f_w and ρ_s . The snow cover model constrains the variability of the DInSAR differential phase as a function of the simulated variability of the physical snow parameters. Moreover, the linear correlation between $\Delta \Phi_{pps}$ and Δh_s is apparent from Fig. 7. This correlation can be used as a precious *a priori* information within the inversion process.

B. SH Estimation Expected Errors

A numerical intercomparison of the snow relative height estimation algorithms can be carried out on simulated data. Fig. 8 shows the comparison between the estimated and "true" SH using the Monte Carlo database for training (50% data) and test (50% data). NNI shows a standard deviation error of about 5.7% (with about zero mean) and an estimate correlation of 0.98. The PAI technique with an average SD of 250 kg/m³ shows a better estimate correlation (about 0.96) than using an extreme value of 500 kg/m³ (0.92, not shown). The two inversion algorithms LRI and MLI appear to be equivalently performing in terms of estimate correlation (about 0.96), mean error, and error standard deviation. These results confirm that the PAI approach is prone to be more biased and inaccurate than other retrieval techniques.

The same analysis can be repeated using a realistic snow scenario, provided by the Alpine3D model, instead of the Monte Carlo random one, as shown in Fig. 9. The PAI algorithm again better performs with an intermediate SD of 200 kg/m³ (estimate bias of -1.4% and estimate correlation of 0.78) than with an upper bound SD of 430 kg/m³ (estimate bias of -1.4% and estimate bias of 0.74. LRI and MLI both show an estimate bias of 0.07% and an estimate correlation of 0.98.

C. SCA Classification

The dry/wet snow classification map, estimated by the proposed SCC-ANN method, is shown in Fig. 10. With the purpose of providing an independent reference, we have performed the same classification using the Nagler-Rott algorithm, described in (13), as shown in Fig. 11. A partial, but noticeable, correspondence for the wet snow areas can be observed by comparing the two maps, with a predominance of wet snow when applying the SCC-ANN retrieval.



Fig. 6. Bi-dimensional histogram of SD and liquid water fraction from Alpine3D snow cover model, using all simulated samples within the area (see Fig. 1) and period of interest (Dec. 9, 2018 till Feb. 25, 2019).



Fig. 7. Same as in Fig. 3 but using liquid water fraction and SD from the Alpine3D snow cover model simulation for the whole period and AOI. (Left plot) Incidence angle $\theta = 20^{\circ}$. (Right plot) $\theta = 40^{\circ}$.



Fig. 8. Snow relative height error estimates using inverse models trained by Monte Carlo random simulated database. Values of "true" SH are Monte Carlo simulated data. (Top-left panel) Using physical analytical inversion (PAI) with $\rho_{ds} = 250 \text{ kg/m}^3$. (Lower-left panel) Using neural network inversion (SPE-ANN). (Top-right panel) Using linear regression inversion (LRI). (Bottom-right panel) Using maximum likelihood inversion (MLI).



Fig. 9. Same as in Fig. 8, but using the Alpine3D snow simulated data.

By applying the NNI technique to data of Figs. 10, 12 shows

the Δh_s map, estimated on Jan. 2, 2019. For the considered couple of dates, a validation has been performed comparing results with the 32 in situ station measurements acquired on Jan. 2, 2019. The comparison indicates a correlation of 0.77

for samples whose DInSAR coherence was found to be above

0.4 (corresponding to a subset of 9 samples on 32 total samples

SCA, estimated from the proposed SCC-ANN method, can

be validated by using MODIS satellite products expressed in

available for that date).



Fig. 10. Example of dry/wet snow cover map estimated by SCC-ANN on Jan. 2, 2019. The AOI is the same as in Fig. 1 and the snow area extent is the one derived from SCC-ANN algorithm.



Fig. 11. Example of dry/wet snow cover map estimated on Jan. 2, 2019 with the algorithm proposed by Nagler and Rott (2016). Magenta color indicates wet snow, whereas cyan indicates dry snow. The AOI is the same as in Fig. 1 and the snow area extent is the one derived from SCC-ANN algorithm.

terms of SCF at about 500-m resolution. This intercomparison is quite cumbersome since MODIS-derived snow cover, even though post-processed to avoid cloud coverage, shows some residual errors that can lead to a significant underestimation of the SCA. These effects may be dependent on the SCF percentage threshold.



Fig. 12. Example of the snow relative height map, estimated by the NNI technique (SPE-ANN) on Jan. 2, 2019.

Within these limitations, we have performed an intercomparison between MODIS SCF and SCC-ANN SCA estimates using 50% of the available dataset. Apart from probability of detection (POD) and false alarm ratio (FAR), we can also use the classification accuracy *ACC* derived from the confusion (or contingency) matrix elements, as follows [60]:

$$ACC = \frac{TN + TP}{TN + TP + FN + FP}$$
(25)

where TN represents the number of True Negatives, TP represents the number of True Positives, FN represents the number of False Negatives, and FP represents the number of False Positives.

Using a MODIS SCF threshold of 1% as a reference, numerical results show that, for the SCC-ANN neural network approach, POD is 0.451, FAR is 0.04 and ACC is 0.78. When increasing the MODIS SCF threshold to 20% and 50%, POD becomes, respectively, 0.55 and 0.60, FAR is 0.11 and 0.24 and ACC 0.83 and 0.87. As mentioned, the POD should be read with caution, but the accuracy is quite high and the FAR is relatively low.

D. SH and Density Estimation

Considering the whole period from Dec. 9, 2018 till Feb. 25, 2019, we have at disposal 211 in situ measurements of SHs and densities within the AOI, shown in Fig. 1. In order to validate results, the SCC-ANN and SPE-ANN neural network retrieval methods can be applied to retrieve Δh_s and ρ_s in correspondence of each ground snow-measurement station.

To quantify the comparison, 4 further error statistical indexes are used to evaluate the overall performance during the considered season. The estimate error can be defined as

$$\delta(x, y, t) = \Delta h_{sm}(x, y, t) - \Delta h_s(x, y, t)$$
(26)

where $\Delta h_{sm}(x, y, t)$ and Δh_s are the measured and estimated relative SHs, respectively, at the location (*x*,*y*) and time *t*. The error metric consists of the error bias b_{δ} , the root mean square



Fig. 13. Bar diagram of error indexes (error bias, RMSE, correlation, and IA) for the snow relative height estimation using the different inversion methods: PAI, LRI, MLI-MC, MLI-A3, MLI-A3C, and SPE-ANN using in situ snow station "true" data between December 9, 2018 and February 25, 2019.

Snow density inversion performances - Central Italy validation 09.12.2018-25.02-2019



Fig. 14. Same as in Fig. 13, but for the SD ρ_s estimation using MLI-MC, MLI-A3, and SPE-ANN.

error (RMSE) r_{δ} and the correlation coefficient ρ_{δ} , defined as

$$\begin{cases}
b_{\delta} = \langle \delta(x, y, t) \rangle \\
r_{\delta} = \langle [\delta(x, y, t)]^{2} \rangle \\
\rho_{\delta} = \langle \Delta h_{sm}(x, y, t) \Delta \hat{h}_{s}(x, y, t) \rangle / \sigma_{hm} \sigma_{h}
\end{cases}$$
(27)

where the angle brackets represent the ensemble averaging over time (number of interferograms) and space (snow measurement sites), where σ_{hm} and σ_h are the standard deviations of Δh_{sm} and $\Delta \hat{h}_s$, respectively. Moreover, the index of agreement (IA) parameter can be useful to provide a comprehensive assessment of the degree of similarity between estimates and measurements,



Fig. 15. Intercomparison of ANN-based snow retrieval with in situ data. (Top panel) Average time-cumulated SH as a function of available overpass dates (see Table II), obtained as a mean of all available in situ data and from all the snow-height retrieval algorithms compared in Fig. 13 (i.e., PAI, LRI, MLI-MC, MLI-A3, MLI-A3C, and SPE-ANN). (Bottom panel) Average snow densities as a function of available overpass dates, obtained as a mean of all available in situ data and from all the snow-density retrieval algorithms compared in Fig. 14 (i.e., MLI-MC, MLI-A3, and SPE-ANN).

and is defined as [64]

$$i_{\delta} = \frac{\left\langle \left[\Delta \hat{h}_{s}\left(x,y,t\right) - \Delta h_{sm}\left(x,y,t\right) \right] \right\rangle^{2}}{\left\langle \left(\left| \Delta \hat{h}_{s}\left(x,y,t\right) - \left\langle \Delta h_{sm}\left(x,y,t\right) \right\rangle \right| + \left| \Delta h_{sm}\left(x,y,t\right) \right| \right) \right\rangle^{2}} \right\rangle^{2}}$$
(28)

being $0 \le i_{\delta} \le 1$ where 0 indicates no agreement and 1 indicates complete match.

Using the in situ validation set, Fig. 13 shows a bar diagram of the performance indexes b_{δ} , r_{δ} , ρ_{δ} , and i_{δ} for the Δh_s estimation, according to different inversion algorithms. The diagram includes six of the inversion approaches, described in Section III and compared in this study, that is, SPE-ANN (neural network) against PAI (physical analytical), LRI (linear regression), MLI-MC (maximum likelihood trained with Monte Carlo), MLI-A3 (maximum likelihood trained with Alpine3D), and MLI-A3C (maximum likelihood trained with Alpine3D and combined with Alpine3D).

Fig. 13 shows that, for the h_s retrieval, the NNI estimation method provides the best performances for all considered four error indexes with a bias of about 0.7 cm, RMSE of about 17 cm, correlation of about 0.5, and IA of about 0.6. The second-best approach is the combined MLI-A3C followed by MLI-A3 and MLI-MC with the PAI inversion algorithm showing the worst estimation capability with a bias of about -2 cm, RMSE of about 30 cm, correlation of about 0.05, and IA of about 0.4. The behavior of PAI is indeed expected to be the worst due to the assumptions that have to be necessarily made on the SD, as discussed previously.

Results for the SD are similarly shown in Fig. 14, even though only three algorithms, providing SD retrieval, are compared (namely, MLI-MC, MLI-A3, SPE-ANN). For the SD estimate, the results obtained from the SPE-ANN neural network approach are again better than maximum-likelihood, with an IA of about 0.85.

Fig. 15 finally shows the average time-cumulated snow as a function of available overpass dates, obtained as a mean of all available in situ data and from all the SH and density retrieval algorithms compared in Figs. 13 and 14. Relative SH retrievals

are converted into snow absolute height by starting from the zero value at the initial date on December 9, 2018. Standard deviations of all estimates at each site at a given date for SH and density are also shown in Fig. 15.

These plots visually confirm what has already been discussed about the capability of ANN-based approaches to better follow the accumulation and ablation of the snow deposit due to snowfall and melting and/or mobilization factors, respectively. In terms of estimate correlation with in situ measurements, the SPE-ANN technique shows a value equal to 0.49 using all 211 total available samples (i.e., when no subselection based on a coherence threshold is performed) that becomes about 0.55 for the sample subset whose DinSAR measured coherence is above 0.4 (corresponding to a subset of 42 samples on 211 samples totally available).

V. CONCLUSION

This work has been devoted to developing a new approach for the estimation of snow cover extent, SH, and SWE, which combines C-band SAR Sentinel-1 satellite data coupled with numerical snow cover model data using ANNs. The AOI is within the Italian central Apennines, which is novel to this kind of studies. Besides the SAR C-band data, auxiliary data sources adopted for this study include optical sensor data (MODIS), atmospheric correction data (GACOS), land cover data (CORINE), and DEMs.

An ANN-based approach has been developed for both wet and dry SCA classification and estimation. Such approach has been compared with other physical and statistical inversion approaches. For the SCA classification, the proposed ANNbased method has been combined with a fuzzy-logic scheme to discriminate between dry and wet snow. For SH and density estimation, the proposed ANN-based method has been compared with a physical-analytical method, linear regressionbased methods, and maximum likelihood methods. Synthetic forward model simulations obtained using a uniform probabilistic distribution (Monte Carlo database), as well as using snow cover model simulations (Alpine3D database), have been used for training purposes in the SH and density estimation methods.

The AOI, in the Italian central Apennines, has been analyzed with a spatial resolution of 100 m using three months of the 2018–2019 winter. SCA retrieval has been validated using MODIS satellite snow cover maps, obtaining accuracies from 0.78 up to 0.87 when discriminating between snow-covered areas and snow-free areas. Maps of dry and wet snow cover obtained from the SCC-ANN have been compared with analogous maps obtained from the Nagler-Rott classification methodology, noting a substantial similarity in SCAs, with more wet snow subareas detected when using the SCC-ANN retrieval. SH estimates have been validated using measurements from more than 200 in situ observations. Imposing a DInSAR coherence larger than 0.4 (corresponding to 42 samples out of a total of 211 samples), we have obtained an estimate correlation value of 0.55. The ANN-based overall error has a bias and a root mean square equal to about 0.5 and 20 cm for SH and to 5 and 80 kg/m³ for SD. Worse results mainly depend on the low DInSAR coherence values, as well as on snow melting effects and meteorologically driven changes of snow state and accumulation.

The main advantages of a DInSAR-based retrieval of snow cover are the achievable spatial resolution, the availability of data in all weather conditions, and the capability of snow layer probing (depending on the SAR frequency). On the other side, the main DInSAR drawbacks consist of relatively small swaths and long revisit periods, causing a coherence loss in the interferograms and, ultimately, unreliable estimates.

However, the growing interest in SAR satellite missions and the current trend in deploying increasingly large satellite constellations could gradually reduce this limitation up to acceptable levels for snow cover retrieval applications.

Future developments of this work will include the extension of the proposed approach to other SAR missions, possibly not only at C band, but also at L and X band. By adopting a multifrequency and multimission SAR approach, we might overcome, totally or in part, some of the limitations we encountered in the present work.

Appendix

A. Snow Permittivity Models

Snow relative permittivity can be calculated directly from dry SD for the case of dry snow. A well-accepted model for dry snow relative permittivity real part ε_{Rrds} using only dry SD ρ_{ds} is expressed by, [27], [55], [56]

$$\varepsilon_{Rrds} = \begin{cases} 1 + a_1 \rho_{ds} + a_2 \rho_{ds}^3 & \rho_{ds} \le \rho_0 \\ \left[(1 - r_{ice}) a_3^{1/3} + r_{ice} a_4^{1/3} \right]^3 & \rho_{ds} > \rho_0 \end{cases}$$
(A.1)

where $r_{ice} = \rho_{ds}/\rho_{ice}$, $a_1 = 1.5995 \text{ cm}^3/\text{g}$, $a_2 = 1.861 \text{ cm}^9/\text{g}$, $a_3 = 1.005$, $a_4 = 3.179$, and $\rho_{ice} = 0.917 \text{ g/cm}^3$. The dry density threshold ρ_0 is equal to 400 kg/m³ [note that in (A.1) ρ_{ds} is expressed in g/cm³].

In general, especially during the melting periods, snow is not dry due to the presence of a liquid water fraction. A simplified frequency-dependent model of snow permittivity is based on the linear mixing (LM) approach that, for the (wet) snow, can be formulated as follows [12]:

$$\varepsilon_{Rrs}^{(LM)}(f) = \varepsilon_{Rrds} \ (1 - f_w) + \varepsilon_{Rrw}(f) f_w \qquad (A.2)$$

where f_w is the volumetric liquid water fraction (%), f is the frequency, and ε_{Rrw} is the liquid water relative permittivity following the Debye model [13]. At C band, the pure water value of ε_{Rrw} is about 66.7 (see later).

The LM model of ε_{Rrs} can be compared with the one from Wiesmann and Mätzler (WM) whose general formulation applies also in wet snow conditions. The WM model exploits the water relaxation spectrum so that the frequency-dependent snow relative permittivity ε_{Rrs} , for small values of f_w , can be decomposed in four additive terms, one of them being the expression for dry snow in (A.1), that is [55]

$$\varepsilon_{Rrs}^{(WM)}(f) = \varepsilon_{Rrds} + Re\left\{\varepsilon_{Rra}\left(f\right) + \varepsilon_{Rrb}\left(f\right) + \varepsilon_{Rrc}\left(f\right)\right\}$$
(A.3)

being *Re* the real part operator with ε_{Rra} , ε_{Rrb} , and ε_{Rrc} the Debye frequency-dependent complex terms. The latter ε_{Rrk} (k = a,b,c) is expressed by

$$\varepsilon_{Rrk} (f) = \varepsilon_{\infty k} + \frac{\varepsilon_{sk} - \varepsilon_{\infty k}}{1 + (f/f_{0k})^2}$$
 (A.4)

where the Debye parameters $\varepsilon_{\infty k}$, ε_{sk} , and f_{0k} are depending on the depolarization factors. Details can be found in (53)–(57) of the WM-cited paper [55].

Another well-known frequency-dependent snow permittivity model is the one proposed by Sihvola and Tiuri (ST) [58], [59]. Based on measurements around 1 GHz scaled to higher frequencies by using the Debye formulas, they proposed for the snow relative permittivity real part ε_{Rrs} the following formula [59]:

 $\varepsilon_{Rrs}^{(ST)}(f) = \varepsilon_{Rrds}^{(ST)} + \Delta \varepsilon_{Rrw}(f_w, f)$

where

$$\begin{cases} \varepsilon_{Rrds}^{(ST)} = c_0 + c_1 \rho_{ds} + c_2 \rho_{ds}^2 \\ \Delta \varepsilon_{Rrw} (f) = (c_3 f_w + c_4 f_w^2) \varepsilon_{Rrw} (f) \\ \varepsilon_{Rrw} (f) = w_0 + \frac{w_1}{1 + (f/f_0)^2} \end{cases}$$
(A.6)

(A.5)

being $c_0 = 1$, $c_1 = 1.7$, $c_2 = 0.7$, $c_3 = 0.1$, $c_4 = 0.8$ with $w_0 = 4.9$, $w_1 = 82.8$, and $f_0 = 8.84$ GHz related to a relaxation time of 18 ps [note that in (A.6) ρ_{ds} is expressed in g/cm³ and f in GHz].

Fig. 16 shows a comparison between the LM simplified model and the more accurate WM and ST models for an average (250 kg/m³) and higher (500 kg/m³) dry SD. All models show a substantial linear dependence on small values of f_w (between 0 and 4%), being the LM providing values of about 60% larger than WM and ST for water fraction larger than 3%. For low values of f_w (i.e., <1%), the overestimation of LM relative permittivity reduces to less than 25%.

To evaluate the WM and ST approximations of snow relative permittivity, we can use the Alpine3D snow cover model simulations which provide, among other snow parameters, both



Fig. 16. Comparison of snow relative permittivity versus snow liquid water fraction using the LM, WM, and ST snow permittivity models. (Left panel) Using dry SD equal to $\rho_{ds} = 250$ kg/m³. (Right panel) Using dry SD equal to $\rho_{ds} = 500$ kg/m³.



Fig. 17. Comparison of snow relative permittivity real part ε_{Rrs} from the ST model versus the WM model using Alpine3D simulations as inputs (see Section II-C). Their discrepancy average and standard deviation are 0.048 and 0.098, respectively.

dry SD and liquid water fraction. Fig. 17 shows the correlation between $\varepsilon_{Rrs}^{(ST)}$ and $\varepsilon_{Rrs}^{(WM)}$ using the Alpine3D outputs for the area and period of interest. In this realistic scenario, dry SD values range from 50 to 400 kg/m³, whereas the water fraction ranges from 0% to 5% for most samples. The discrepancy average and standard deviation between the two snow permittivity models in (A.3) and (A.5) are 0.048 and 0.098, respectively, that is ST and WM models differ less than 3% on average. The overestimation of LM permittivity model with respect to WM and ST models is (not shown) 0.28 and 0.33 on average, respectively, that is about 16% and 18% on average.

B. Differential Phase Uncertainty Analysis

Starting from (5), we can use the LM snow permittivity model for a dry SD less than 400 kg/m^3 in (A.3) to derive the following

expression for the DInSAR differential phase, in the case of dry snow, for $\theta_t = 0^{\circ}$

$$\Delta \Phi_{pps} = f \frac{4\pi \Delta R}{v_s} = -2k \left(\cos \theta_l - \sqrt{\varepsilon_{Rrs} - \sin^2 \theta_l} \right) \Delta h_s.$$
(B.1)

For a wet snow condition, we can employ the LM model expressed in (A.2), a choice that may represent an upper bound of snow permittivity values (see Fig. 16).

We can then generalize the previous expression for dry snow as

$$\Delta \Phi_{pps} = f \frac{4\pi\Delta R}{v_s} = -2k \left(\cos \theta_l - \sqrt{\left(1 + a_1\rho_{ds} + a_2\rho_{ds}^3\right)\left(1 - f_w\right) + \varepsilon_{Rrw}f_w - \sin^2\theta_l}\right) \Delta h_s.$$
(B.2)

Using (B.2), the partial derivatives of the three terms of (9) in Section II-C are the following:

$$\begin{pmatrix}
\frac{\partial \Delta \Phi_{pps}}{\partial \theta_l} = -2k\Delta h_s \\
\times \left(\frac{\sin(\theta)\cos(\theta)}{\sqrt{(1-f_w)(1+a_1\rho_{ds}+a_2\rho_{ds}^3) + \varepsilon_{Rrw}f_w - \sin^2(\theta)}} - \sin(\theta) \right) \\
\frac{\partial \Delta \Phi_{pps}}{\partial \rho_{ds}} = k\Delta h_s \\
\times \left(\frac{(1-f_w)(a_1+3a_2\rho_{ds}^2)}{\sqrt{(1-f_w)(1+a_1\rho_{ds}+a_2\rho_{ds}^3) + \varepsilon_{Rrw}f_w - \sin^2(\theta)}}} \right) \\
\frac{\partial \Delta \Phi_{pps}}{\partial f_w} = -k\Delta h_s \\
\times \left(\frac{1+a_1\rho_{ds}+a_2\rho_{ds}^3 - \varepsilon_{Rrw}}{\sqrt{(1-f_w)(1+a_1\rho_{ds}+a_2\rho_{ds}^3) + \varepsilon_{Rrw}f_w - \sin^2(\theta)}}} \right) \\
(B.3)$$

Substituting (B.3) into (9), we get the $\Delta \Phi_{pps}$ uncertainty expression as a function of the key parameters θ_l , ρ_{ds} , and f_w .

Fig. 18 shows the sensitivity of $\Delta \Phi_{pps}$ as a function of Δh_s for lower and higher values of θ_l , f_w , and ρ_{ds} . The curves are linear



Fig. 18. Uncertainty of DInSAR differential phase $\Delta \Phi_{pps}$ versus snow relative height, using (9) varying the liquid water fraction (top panels with low value $f_w = 0.1\%$, bottom panels with large value $f_w = 4\%$) and local incidence angle (left panels with lower value $\theta_l = 20^\circ$, right panels with higher value $\theta_l = 40^\circ$). In (9), the standard deviations are set to $\sigma_{\theta_l} = 3^\circ$, $\sigma_{f_w} = 0.1 f_w$, $\sigma_{\rho_{ds}} = 100 \text{ kg/m}^3$.

and symmetric with respect to Δh_s due to its proportionality from (B.3). For a given incidence angle, a larger liquid water fraction increases the differential phase uncertainty, especially for lower SD. For a given SD, the differential phase is larger for larger incidence angles. For larger water fraction, the effect of the dry SD can be considered negligible, especially for smaller snow thickness. The largest uncertainty of the differential phase is expected for highly wet deep snow layers and larger incidence angles.

C. Synoptic View of Winter 2018–2019

Fig. 19 shows that winter 2018–2019 was not abnormal compared to 1981–2010 mean.

Indeed, during December 2018, the Central Apennines were mainly influenced by a high-pressure system which caused a positive temperature anomaly and a deficit of precipitations. January 2019 instead was dominated by a low-pressure system which brought low temperatures and a positive precipitation anomaly. During February, again, a high-pressure system caused a positive temperature anomaly and a small deficit of precipitations. Thus, monthly anomalies compensate each other and winter 2018–2019 results to be in line with the average of past winters.

D. Estimating Mean Snowpack Density From Snowpack Top Layer Density

The density data collected daily by the Meteomont service unfortunately refers only to the snowpack top layer, thus they are not representative of the mean snowpack density.

To overcome this problem and to be able to use the snowpack top density observations for validation, we used an independent dataset of snowpack vertical profiles, still provided by the Meteomont service, to find a relation between the top and mean snowpack density values.

The new dataset consists of 138 vertical stratigraphies of the snowpack properties done during winters 2018–2019, 2019–2020, and 2020–2021, which include the SD of each layer identified in the snowpack. The stratigraphies are done weekly, and since just a few of them occurred on the same day of SAR acquisitions, we could not use them for validation.

Thus, for each vertical profile, we calculated the mean density and we extracted the density of the top layer. Then we compared top and mean density obtaining a correlation coefficient equal



Fig. 19. Synoptic view of the situation of winter 2018–2019 is shown throughout the 12 panels. The columns of the figure represent the 500 hPa geopotential height anomaly, the 850 hPa temperature anomaly, and the surface precipitation anomaly. The rows represent the December-to-February composite anomaly, December anomaly, January anomaly, and February anomaly, respectively.

to 0.7 (see Fig. 20), and we found the following linear relation:

$$\rho_{\rm mean} = 1.1 \rho_{\rm top} + 55.5$$
 (D.1)

where $\rho_{\rm mean}$ and $\rho_{\rm top}$ are the mean and the top snowpack densities, respectively.

Substituting to ρ_{top} the density value coming from the daily Meteomont dataset, we were able to estimate the corresponding ρ_{mean} , and use it to validate the ANN estimations.

ACKNOWLEDGMENT

The authors wish to thank the ESA Research Service and Support (RSS) group from Frascati (Rome, Italy) for the valuable help in improving the SAR processing chain. The European Space Agency and its Network of Resources (NoR) are also acknowledged for sponsoring and providing high-performance computing resources. We are grateful to the Italian Meteomont Avalanche Warning Service (MAWS)



Fig. 20. Figure shows the comparison between the top and mean snowpack density of 138 vertical stratigraphies of the snowpack properties done during winters 2018–2019, 2019–2020, and 2020–2021 by the Meteomont service.

(https://meteomont.carabinieri.it/home) for providing daily snow data within the area of interest. The authors wish to commemorate one of the co-authors of this manuscript, Prof. Frank Silvio Marzano, who suddenly passed away on May 2022 after the research work presented herein had already been completed. His support was fundamental in the organization of the study and in the writing of this article. His contribution was—and still is—inspiring in many ways.

REFERENCES

- P. Lemke et al., "Observations: Changes in snow, ice and frozen ground," in Climate Change 2007: The Physical Science Basis. Contribution of Working Group 1 to the Fourth Assessment Report of the Intergovernmental Panel On Climate Change. Cambridge, U.K.: Cambridge Univ. Press, 2007.
- [2] M. Sturm, M. A. Goldstein, and C. Parr, "Water and life from snow: A trillion-dollar science question," *Water Resour. Res.*, vol. 53, pp. 3534–3544, 2017, doi: 10.1002/2017WR020840.
- [3] P. Krajči, M. Danko, Z. Kostka, and L. Holko, "Experimental measurements for improved understanding and simulation of snowmelt events in the Western Tatra Mountains," *J. Hydrol. Hydromech.*, vol. 64, no. 4, pp. 316–328, 2016.
- [4] A. Frei et al., "A review of global satellite-derived snow products," *Adv. Space Res.*, vol. 50, no. 8, pp. 1007–1029, 2012, doi: 10.1016/ j.asr.2011.12.021.
- [5] S. Jawak, T. Bidawe, and A. Luis, "A review on applications of imaging synthetic aperture radar with a special focus on cryospheric studies," *Adv. Remote Sens.*, vol. 4, pp. 163–175, 2015, doi: 10.4236/ars.2015.42014.
- [6] Y.-L. S. Tsai, A. Dietz, N. Oppelt, and C. Kuenzer, "Remote sensing of snow cover using spaceborne SAR: A review," *Remote Sens.*, vol. 11, 2019, Art. no. 1456, doi: 10.3390/rs11121456.
- [7] T. Strozzi, U. Wegmuller, and C. Mätzler, "Mapping wet snowcovers with SAR interferometry," *Int. J. Remote Sens.*, vol. 20, pp. 2395–2403, 1999.
- [8] J. Shi and J. Dozier, "Estimation of snow water equivalence using SIR-C/X-SAR. I. Inferring snow density and subsurface properties," *IEEE Trans. Geosci. Remote Sens.*, vol. 38, no. 6, pp. 2465–2474, Nov. 2000.
- [9] J. Shi and J. Dozier, "Estimation of snow water equivalence using SIR-C/X-SAR. II. Inferring snow depth and particle size," *IEEE Trans. Geosci. Remote Sens.*, vol. 38, no. 6, pp. 2475–2488, Nov. 2000.
- [10] T. Nagler and H. Rott, "Retrieval of wet snow by means of multitemporal SAR data," *IEEE Trans. Geosci. Remote Sens.*, vol. 38, no. 2, pp. 754–765, Mar. 2000.
- [11] T. Guneriussen, K. Hogda, H. Johnsen, and I. Lauknes, "InSAR for estimation of changes in snow water equivalent of dry snow," *IEEE Trans. Geosci. Remote Sens.*, vol. 39, no. 10, pp. 2101–2108, Oct. 2001.

- [12] F. Ulaby and D. G. Long, *Microwave Radar and Radiometric Remote Sensing*. Boston, MA, USA: Artech House, 2014.
- [13] D. Solimini, Understanding Earth Observation: The Electromagnetic Foundation of Remote Sensing. London, U.K.: Springer, 2016.
- [14] B. Guan, N. P. Molotch, D. E. Waliser, S. M. Jepsen, T. H. Painter, and J. Dozier, "Snow water equivalent in the Sierra Nevada: Blending snow sensor observations with snowmelt model simulations," *Water Resour. Res.*, vol. 49, pp. 5029–5046, 2013.
- [15] C. Notarnicola, R. Ratti, V. Maddalena, T. Schellenberger, B. Ventura, and M. Zebisch, "Seasonal snow cover mapping in Alpine areas through time series of COSMO-SkyMed images," *IEEE Geosci. Remote Sens. Lett.*, vol. 10, no. 4, pp. 716–720, Jul. 2013.
- [16] S. Pettinato, E. Santi, S. Paloscia, B. Aiazzi, S. Baronti, and A Garzelli, "Snow cover area identification by using a change detection method applied to COSMO-SkyMed images," *J. Appl. Remote Sens.*, vol. 8, 2014, Art. no. 084684, doi: 10.1117/1.JRS.8.
- [17] L. Pulvirenti, F. S. Marzano, N. Pierdicca, S. Mori, and M. Chini, "Discrimination of water surfaces, heavy rainfall, and wet snow using COSMO-SkyMed observations of severe weather events," *IEEE Trans. Geosci. Remote Sens.*, vol. 52, no. 2, pp. 858–869, Feb. 2014.
- [18] Y. Wang, L. Wang, H. Li, Y. Yang, and T. Yang, "Assessment of snow status changes using L-HH temporal-coherence components at Mt. Dagu, China," *Remote Sens.*, vol. 7, no. 9, pp. 11602–11620, 2015.
- [19] S. Shaobo et al., "Estimation and analysis of snow water equivalents based on C-band SAR data and field measurements," *Arctic, Antarctic, Alpine Res.*, vol. 47, no. 2, pp. 313–326, 2015, doi: 10.1657/AAAR00C-13-135.
- [20] A. Muhuri, S. Manickam, and A. Bhattacharya, "Scattering mechanism based snow cover mapping using RADARSAT-2 C-band polarimetric SAR data," *IEEE J. Sel. Topics Appl. Earth Observ. Remote Sens.*, vol. 10, no. 7, pp. 3213–3224, Jul. 2017.
- [21] E. Santi et al., "Exploiting the ANN potential in estimating snow depth and snow water equivalent from the airborne SnowSAR data at Xand Ku-bands," *IEEE Trans. Geosci. Remote Sens.*, vol. 60, Jun. 2021, Art. no. 4301216, doi: 10.1109/tgrs.2021.3086893.
- [22] F. M. Bianchi, J. Grahn, M. Eckerstorfer, E. Malnes, and H. Vickers, "Snow avalanche segmentation in SAR images with fully convolutional neural networks," *IEEE J. Sel. Topics Appl. Earth Observ. Remote Sens.*, vol. 14, pp. 75–82, Nov. 2021, doi: 10.1109/JSTARS. 2020.3036914.
- [23] M. Proksch et al., "MEMLS3&a: Microwave emission model of layered snowpacks adapted to include backscattering," *Geosci. Model Dev.*, vol. 8, pp. 2611–2626, 2015, doi: 10.5194/gmd-8-2611-2015.
- [24] G. Picard, M. Sandells, and H. Löwe, "SMRT: An active–passive microwave radiative transfer model for snow with multiple microstructure and scattering formulations (v1.0)," *Geosci. Model Develop.*, vol. 11, pp. 2763–2788, 2018, doi: 10.5194/gmd-11-2763-2018.
- [25] E. J. Deeb, R. R. Forster, and D. L. Kane, "Monitoring snow cover evolution using interferometric synthetic aperture radar on the north slope of Alaska, USA," *Int. J. Remote Sens.*, vol. 32, no. 14, pp. 3985–4003, 2011.
- [26] S. Leinss, G. Parrella, and I. Hajnsek, "Snow height determination by polarimetric phase differences in X-band SAR data," *IEEE J. Sel. Topics Appl. Earth Observ. Remote Sens.*, vol. 7, no. 9, pp. 3794–3810, Sep. 2014, doi: 10.1109/JSTARS.2014.2323199.
- [27] S. Leinss, A. Wiesmann, J. Lemmetyinen, and I. Hajnsek, "Snow water equivalent of dry snow measured by differential interferometry," *IEEE J. Sel. Topics Appl. Earth Observ. Remote Sens.*, vol. 8, no. 8, pp. 3773–3790, Aug. 2015.
- [28] T. Nagler, H. Rott, E. Ripper, G. Bippus, and M. Hetzenecker, "Advancements for snowmelt monitoring by means of Sentinel-1 SAR," *Remote Sens.*, vol. 8, 2016, Art. no. 348.
- [29] H. Li, Z. Wang, G. He, and W. Man, "Estimating snow depth and snow water equivalence using repeat-pass interferometric SAR in the Northern Piedmont region of the Tianshan Mountains," *J. Sensors*, vol. 2017, 2017, Art. no. 8739598, doi: 10.1155/2017/8739598.
- [30] V. Conde, G. Nico, P. Mateus, J. Catalão, A. Kontu, and M. Gritsevich, "On the estimation of temporal changes of snow water equivalent by spaceborne sar interferometry: A new application for the Sentinel-1 mission," *J. Hydrol. Hydromech.*, vol. 67, no. 1, pp. 93–100, 2019, doi: 10.2478/johh-2018-0003.
- [31] Y. L. S. Tsai, A. Dietz, N. Oppelt, and C. Kuenzer, "Wet and dry snow detection using Sentinel-1 SAR data for mountainous areas with a machine learning technique," *Remote Sens.*, vol. 11, 2019, Art. no. 895.
- [32] A. B. Mahmoodzada, D. Varade, and S. Shimada, "Estimation of snow depth in the Hindu Kush Himalayas of Afghanistan during peak winter and early melt season," *Remote Sens.*, vol. 12, no. 17, 2020, Art. no. 2788.

- [33] C. J. Crawford, S. M. Manson, M. E. Bauer, and D. K. Hall, "Multitemporal snow cover mapping in mountainous terrain for Landsat climate data record development," *Remote Sens. Environ.*, vol. 135, pp. 224–233, 2013.
- [34] H. Guangjun, X. Pengfeng, F. Xuezhi, Z. Xueliang, W. Zuo, and C. Ni, "Extracting snow cover in mountain areas based on SAR and optical data," *IEEE Geosci. Remote Sens. Lett.*, vol. 12, no. 5, pp. 1136–1140, May 2015.
- [35] G. He et al., "Dry and wet snow cover mapping in mountain areas using SAR and optical remote sensing data," *IEEE J. Sel. Topics Appl. Earth Observ. Remote Sens.*, vol. 10, no. 6, pp. 2575–2588, Jun. 2017.
- [36] B. Snapir, A. Momblanch, S. Jain, T. Waine, and I. Holman, "A method for monthly mapping of wet and dry snow using Sentinel-1 and MODIS: Application to a Himalayan River basin," *Int. J. Appl. Earth Observ. Geoinf.*, vol. 74, pp. 222–230, 2019.
- [37] S. Leinss, G. Parrella, and I. Hajnsek, "Snow height determination by polarimetric phase differences in X-band SAR data," *IEEE J. Sel. Topics Appl. Earth Observ. Remote Sens.*, vol. 7, no. 9, pp. 3794–3810, Sep. 2014, doi: 10.1109/JSTARS.2014.2323199.
- [38] S.-E. Park, Y. Yamaguchi, G. Singh, S. Yamaguchi, and A. C. Whitaker, "Polarimetric SAR response of snow-covered area observed by multitemporal ALOS PALSAR fully polarimetric mode," *IEEE Trans. Geosci. Remote Sens.*, vol. 52, no. 1, pp. 329–340, Jan. 2014.
- [39] A. Muhuri, D. Ratha, and A. Bhattacharya, "Seasonal snow cover change detection over the Indian Himalayas using polarimetric SAR images," *IEEE Geosci. Remote Sens. Lett.*, vol. 14, no. 12, pp. 2340–2344, Dec. 2017.
- [40] D. Varade, A. K. Maurya, O. Dikshit, G. Singh, and S. Manickam, "Snow depth in Dhundi: An estimate based on weighted bias corrected differential phase observations of dual polarimetric bi-temporal Sentinel-1 data," *Int. J. Remote Sens.*, vol. 4, pp. 3031–3053, 2020, doi: 10.1080/01431161.2019.1698076.
- [41] A. Patil, G. Singh, C. Rüdiger, S. Mohanty, S. Kumar, and S. Mani, "A novel approach for the snow water equivalent retrieval using X-band polarimetric synthetic aperture radar data," *IEEE Trans. Geosci. Remote Sens.*, vol. 59, no. 5, pp. 3753–3763, May 2021, doi: 10.1109/TGRS.2020.3016527.
- [42] J. Lemmetyinen et al., "Retrieval of effective correlation length and snow water equivalent from radar and passive microwave measurements," *Remote Sens.*, vol. 10, no. 2, 2018, Art. no. 170, doi: 10.3390/rs10020170.
- [43] J. Pulliainen, "Mapping of snow water equivalent and snow depth in boreal and sub-arctic zones by assimilating space-borne microwave radiometer data and ground-based observations," *Remote Sens. Environ.*, vol. 101, pp. 257–269, 2006.
- [44] E. Raparelli, P. Tuccella, V. Colaiuda, and F. S. Marzano, "Snow cover prediction in the Italian central Apennines using weather forecast and land surface numerical models," *Cryosphere*, vol. 17, pp. 519–538, 2023, doi: 10.5194/tc-17-519-2023.
- [45] F. Pedrotti and D. Gafta, "The high Mountain Flora and vegetation of the Apennines and the Italian Alps," in *Alpine Biodiversity in Europe. Ecological Studies*, L. Nagy, G. Grabherr, C. Körner, and D. B. A. Thompson, Eds. New York, NY, USA: Springer, 2003, pp. 73–84.
- [46] H. Luijting, D. Vikhamar-Schuler, T. Aspelien, Å. Bakketun, and M. Homleid, "Forcing the SURFEX/Crocus snow model with combined hourly meteorological forecasts and gridded observations in southern Norway," *Cryosphere*, vol. 12, pp. 2123–2145, 2018, doi: 10.5194/tc-12-2123-2018.
- [47] S. Morin et al., "Application of physical snow cover models in support of operational avalanche hazard forecasting: A status report on current implementations and prospects for the future," *Cold Regions Sci. Technol.*, vol. 170, 2019, Art. no. 102910, doi: 10.1016/j.coldregions.2019.102910.
- [48] N. Longepe, S. Allain, L. Ferro-Famil, E. Pottier, and Y. Durand, "Snowpack characterization in mountainous regions using C-band SAR data and a meteorological model," *IEEE Trans. Geosci. Remote Sens.*, vol. 47, no. 2, pp. 406–418, Feb. 2009.
- [49] H. A. Zebker, P. A. Rosen, and S. Hensley, "Atmospheric effects in interferometric synthetic aperture radar surface deformation and topographic maps," J. Geophys. Res., vol. 102, pp. 7547–7563, 1997.
- [50] G. Nico, R. Tome, J. Catalão, and P. M. Miranda, "On the use of the WRF model to mitigate tropospheric phase delay effects in SAR interferograms," *IEEE Trans. Geosci. Remote Sens.*, vol. 49, no. 12, pp. 4970–4976, Dec. 2011.
- [51] E. Pichelli et al., "InSAR water vapor data assimilation into mesoscale model MM5: Technique and pilot study," *IEEE J. Sel. Topics Appl. Earth Observ. Remote Sens.*, vol. 8, no. 8, pp. 3859–3875, Aug. 2015, doi: 10.1109/JSTARS.2014.2357685.

- [52] P. Mateus, G. Nico, and J. Catalão, "Uncertainty assessment of the estimated atmospheric delay obtained by a numerical weather model (NMW)," *IEEE Trans. Geosci. Remote Sens.*, vol. 53, no. 12, pp. 6710–6717, Dec. 2015.
- [53] C. Yu, Z. Li, N. T. Penna, and P. Crippa, "Generic atmospheric correction model for interferometric synthetic aperture radar observations," J. *Geophys. Res., Solid Earth*, vol. 123, pp. 9202–9222, 2018.
- [54] G. Büttner, B. Kosztra, and G. Maucha, "Accuracy assessment of CLC data," in *European Landscape Dynamics: CORINE Land Cover Data*, J. Feranec, T. Soukup, G. Hazeu, and G. Jaffrain Eds. Boca Raton, FL, USA: CRC Press, 2016, pp. 41–52.
- [55] C. Mätzler, "Microwave permittivity of dry snow," *IEEE Trans. Geosci. Remote Sens.*, vol. 34, no. 2, pp. 573–581, Mar. 1996.
- [56] A. Wiesmann and C. Mätzler, "Microwave emission model of layered snowpacks," *Remote Sens. Environ.*, vol. 70, pp. 307–316, 1999, doi: 10.1016/S0034-4257(99)00046-2.
- [57] C. Mätzler and A. Wiesmann, "Extension of the microwave emission model of layered snowpacks to coarse-grained snow," *Remote Sens. Environ.*, vol. 70, pp. 317–325, 1999, doi: 10.1-016/S0034-4257(99)00047-4.
- [58] A. Sihvola and M. Tiuri, "Snow fork for field determination of the density and wetness profiles of a snowpack," *IEEE Trans. Geosci. Remote Sens.*, vol. GE-24, no. 5, pp. 717–721, Sep. 1986.
- [59] M. Tiuri, A. Sihvola, E. Nyfors, and M. Hallikainen, "The complex dielectric constant of snow at microwave frequencies," *IEEE J. Ocean. Eng.*, vol. 9, no. 5, pp. 377–382, Dec. 1984.
- [60] S. Haykin, *Neural Networks and Learning Machines*. Upper Saddle River, NJ, USA: Pearson Education, 2009.
- [61] F. S. Marzano, A. Mugnai, G. Panegrossi, N. Pierdicca, E. A. Smith, and J. Turk, "Bayesian estimation of precipitating cloud parameters from combined measurements of spaceborne microwave radiometer and radar," *IEEE Trans. Geosci. Remote Sens.*, vol. 37, no. 1, pp. 596–613, Jan. 1999.
- [62] F. S. Marzano, D. Scaranari, and G. Vulpiani, "Supervised fuzzylogic classification of hydrometeors using C-band dual-polarized radars," *IEEE Trans. Geosci. Remote Sens.*, vol. 45, no. 11, pp. 3784–3799, Dec. 2007.
- [63] C. W. Chen and H. A. Zebker, "Phase unwrapping for large SAR interferograms: Statistical segmentation and generalized network models," *IEEE Trans. Geosci. Remote Sens.*, vol. 40, no. 8, pp. 1709–1719, Aug. 2002.
- [64] C. J. Willmott, "On the validation of models," *Phys. Geogr.*, vol. 2, pp. 184–194, 1981.



Gianluca Palermo received the Laurea degree in electronic engineering, the master's specialization degree in satellites and orbiting platforms, and the Ph.D. degree in aerospace engineering from the Sapienza University of Rome, Rome, Italy, in 2013, 2014, and 2018, respectively.

After being a Postdoc with the Mechanical and Aeronautics Department Sapienza University of Rome, working on concurrent engineering and optimization methods applied to space system design. In 2020, he joined the Department of Information

Engineering, Sapienza University of Rome to focus on SAR data processing and snow cover properties retrieval.



Edoardo Raparelli received the Bachelor's degree in physics from the University of L'Aquila, L'Aquila, Italy, in 2015, the Master degree in atmospheric and cryospheric sciences from the University of Innsbruck, Innsbruck, Austria, in 2018, and the Ph.D. degree in information and communication technologies from the La Sapienza University of Rome, Rome, Italy, in 2022.

During 2018, he has been a visiting student with the WSL Institute for Snow and Avalanche Research SLF, Davos, Switzerland, where he studied the ap-

plicability of modern parametrizations for new snow settlement to a real snow cover. While pursuing Ph.D. degree, he investigated the potentials of combining model-simulated and remotely retrieved snow cover data to reproduce the observed snow cover properties.



Paolo Tuccella received the Laurea degree in physics and the Ph.D. degree in physics from the University of L'Aquila, L'Aquila, Italy, in 2009 and 2013, respectively.

As part of the Ph.D., he spent a period of six months with the National Oceanic and Atmospheric Administration, Boulder, CO, USA, in 2012. After six months of postdoc in Italy, in 2013, he began his career as a Postdoc Researcher abroad with the University Pierre et Marie Curie, Paris, France, as part of the European project ACCESS (ArcticClimate

Change, Economy and Society) aimed at studying the impact of emissions of oil and gas extraction on the air quality of the Arctic. Later, in 2015, he worked as Postdoc with the Ecole Polytechnique, Paris, with the task to develop a coupled version of theWRF meteorological model and the chemical transport model CHIMERE. In 2016, he received a two-year AXA Research Fund grant with a project on the risk assessment in the climate-environment field. He is currently a Researcher on tenure track with the University of L'Aquila. The research topic is about the aerosol impact on the climate and modelling the seasonal snow cover.



Since February 2019, he has been working as an EO Research Support Engineer with Progressive Systems S.r.l., Frascati, Italy.



Frank Silvio Marzano (Fellow, IEEE) received the Laurea degree (*cum laude*) in electrical engineering and the Ph.D. degree in applied electromagnetics both from the University of Rome "La Sapienza," Rome, Italy, in 1988 and 1993, respectively.

In 1992, he was a Visiting Scientist with the Florida State University, Tallahassee, FL, USA. During 1993, he collaborated with the Institute of Atmospheric Physics, National Council of Research (CNR), Rome, Italy. From 1994 to 1996, he was with the Italian Space Agency, Rome, Italy, as a post-doctorate re-

searcher. After being a Lecturer at the University of Perugia, Perugia, Italy, in 1997, he joined the Department of Electrical Engineering, University of L'Aquila, L'Aquila, Italy, teaching courses on electromagnetic fields as Assistant Professor. In 1999, he was with the Naval Research Laboratory, Monterey, CA, USA, as a Visiting Scientist. In 2002, he got the qualification to Associate Professorship and has co-founded Center of Excellence on Remote Sensing and Hydro-Meteorological Modeling (CETEMPS), L'Aquila. In 2005, he finally joined the Department of Information Engineering, Electronics and Telecommunications, Sapienza University of Rome, Rome, Italy, where he taught courses on antennas, propagation, and remote sensing. Since 2007, he was Vice-Director of CETEMPS, University of L'Aquila, L'Aquila, Italy, where he was nominated director on March 2013. He has published more than 130 papers on refereed International Journals, more than 30 contributions to international Book chapters, and more than 300 extended abstract on international and national congress proceedings. He was the Editor of two books. His research interests include passive and active remote sensing of the atmosphere from ground-based, airborne, and space-borne platforms and electromagnetic propagation studies.

Dr. Marzano was an Associated Editor of IEEE GEOSCIENCE REMOTE SENS-ING LETTERS (GRSL) from January 2004 to June 2014, and since mid-2014, he was Associated Editor of IEEE TRANSACTIONS ON GEOSCIENCE AND REMOTE SENSING (TGRS). In 2005 and 2007, he has been Guest Co-Editor of the MicroRad04 and MicroRad06 Special Issues for IEEE-TGRS. Since January 2011, he was an Associate Editor of the journal *EGUAtmospheric Measurements Techniques*. He was Fellow of RMetS (Royal Meteorological Society) since 2012.

Open Access funding provided by 'Università degli Studi di Roma "La Sapienza" within the CRUI CARE Agreement